Author's response to reviews

Title: Oesophageal Perforation: A diagnostic and therapeutic challenge in a resource limited setting. A report of three cases

Authors:

Deo D Balumuka (balumukad@gmail.com)
Philipo L Chalya (drphillipoleo@yahoo.com)
William Mahalu (William.mahalu@yahoo.com)

Version: 4 Date: 21 September 2011

Author's response to reviews:

dear editor,,

i have noted the new major changes to make and i have made the changes, as per the submission system. The names of Chalya and Deo have been formatted. all the others are from the same institution.
balumuka deo darius and william mahalu are from the department of cardiothoracic surgery in weill bugando university college of health sciences and philipo chalya is in the department of surgery.

thank you

dr.balumuka darius